Author's response to reviews

Title: A 41 year old male presenting with massive malignant solitary fibrous tumor arising from the bladder serosa: a case report

Authors:

Jordan M Dozier (dozierjm@njms.rutgers.edu)
Zubin M Bamboat (bamboatz@cancersurgeryassociates.com)
Donald A Mccain (drdonaldmccain@aol.com)
Patrice Hassoun (Phassoun@hackensackumc.org)
Zena Jameel (Jameelze@gmail.com)

Version: 4 Date: 17 July 2014

Author's response to reviews: see over
To the Journal of Medical Case Reports,

Enclosed please find our case report of a malignant solitary fibrous tumor arising from the bladder serosa. We discuss the presentation, differential diagnosis, pathologic findings and management of this extremely rare disease and discuss existing literature on this topic. We believe the information provided will be of interest to the *Journal of Medical Case Reports* readership as this is an original case report that describes the largest malignant sarcoma arising from the bladder serosa. Furthermore, we believe that the content of this report will be of particular interest to surgical oncologist as well as pathologist given the unique histologic subtype observed in our patient. We hope you find it acceptable for publication in your journal. The authors have no competing interests.

Sincerely,

Zubin M. Bamboat, MD (Corresponding Author)